Author's response to reviews

Title: Gender and age differences among current smokers in a general population survey

Authors:

Ulrich John (ujohn@uni-greifswald.de)
Monika Hanke (hanke@uni-greifswald.de)
Christian Meyer (chmeyer@uni-greifswald.de)
Anja Schumann (schumann@uni-greifswald.de)

Version: 3  Date: 27 May 2005

Author’s response to reviews:

BMC Public Health
Editor

May 27, 2005

Resubmission of a manuscript
Title: "Gender and age differences among current smokers in a general population survey"
MS ID: 8391026696089199
Authors: Ulrich John, Monika Hanke, Christian Meyer and Anja Schumann
Journal: BMC Public Health

Dear editor,

We are happy that you accepted our paper by your e-mail yesterday. There were no further recommendations for changes in the text from the reviewers. We just changed what you requested (qualifications of the authors, affiliations, word count, keywords) and made the writing of statistical values consistent (w- and p-values on page 5), specified the chi^2 test that we used (Likelihood chi^2 test, page 7), and made the writing "at the age of 40 or above" consistent on page 12.

Corresponding author: Ulrich John, Prof., Director of the Institute of Epidemiology and Social Medicine, University of Greifswald, W.-Rathenau-Str. 48, D-17487 Greifswald, phone +3834 867700, fax +3834 867701
email: ujohn@uni-greifswald.de
internet: http://www.medizin.uni-greifswald.de/epidem

With kind regards

Ulrich John
(on behalf of all authors)